Author's response to reviews

Title: Comparative efficacy of two primary care interventions to assist withdrawal from long term benzodiazepine use: A protocol for a clustered, randomized clinical trial

Authors:

Caterina Vicens (caterinavicens@gmail.com)
Isabel Socias (imsocias@ibsalut.caib.es)
Catalina Mateu (cmateus@ibsalut.caib.es)
Alfonso Leiva (aleiva@ibsalut.caib.es)
Ferran Bejarano (fbejarano.tarte.ics@gencat.cat)
Ermengol Sempere (meresempere@gmail.com)
Josep Basora (jbasora.tarte.ics@gencat.cat)
Vicente Palop (vpalop@hospital-ribera.com)
Marta Mengual (mmengual.tarte.ics@gencat.cat)
Jose Luis Beltran (beltran_jos@gva.es)
Enric Aragonès (earagones.tarte.ics@gencat.cat)
Guillem Lera (guillemlera@comv.es)
Silvia Folch (sfolch.tarte.ics@gencat.cat)
Josep Lluís Piñol (jlpinyl.tarte.ics@gencat.cat)
Magdalena Esteva (mesteva@ibsalut.caib.es)
Miguel Roca (mroca@uib.es)
Arturo Arenas (ajarenas@ibsalut.caib.es)
Maria del Mar Sureda (mmsureda@ibsalut.caib.es)
Francisco Campoamor (francisco.camposamor@ssib.es)
Francisca Fiol (ffiol@ibsalut.caib.es)

Version: 2 Date: 14 April 2011

Author’s response to reviews:

Dear revisor

I will comment the chages done as an answer at your comments

Abstract: This should be identical in the manuscript and the submission system---- It has been done

References: please provide a functioning link to website reference number 4.

-----I will change for this website reference

Tables: we notice that you have included the tables as additional files. If
you want the tables to be visible within the final published manuscript
please include them in the manuscript in a tables section following the
references. Alternatively, please cite the files as Additional file 1 etc.,
and include an additional files section in the manuscript.----- The table has been
included in the manuscript.

It has been done the minor revision: The changes in the
Email address, title of the figure and check for typographical errors.

your sincerely
Caterine Vicens